Author's response to reviews

Title: Predictors of Knowledge of H1N1 Infection and Transmission in the U.S. Population

Authors:

   Elena Savoia (esavoia@hsph.harvard.edu)
   Marcia A Testa (testa@hsph.harvard.edu)
   Kasisomayajula Viswanath (vish_viswanath@dfci.harvard.edu)

Version: 4 Date: 25 April 2012

Author's response to reviews:

Dear editor,

the submitted new version of the manuscript includes the revisions requested:

1. Abstract was inserted in the main manuscript
2. Competing interests, acknowledgments and authors' contributions moved prior to the references
3. Vertical lines and text removed from the tables

Thanks

Elena Savoia

617-384-9055